Author's response to reviews

Title: Mesenteric Panniculitis with pedal edema: A case report

Authors:

Muhammad A Rauf (ameenrauf@gmail.com)
Abdul M Zafar (amueed@gmail.com)
Tabish Chawla (tabish.chawla@aku.edu)
Gul e Khanda (gule.khanda@aku.edu)

Version: 2 Date: 11 May 2008

Author's response to reviews:

Dear Editor,

Greetings,

In view of the editorial comments, revisions of the manuscript have been completed. The details are as follows,

1. Patient age, sex and ethnic origin have been mentioned in the abstract.
2. Abbreviations have been replaced.
3. 'Case report' has been rephrased as 'case presentation'.
4. Patient’s age, sex and ethnic origin has been added in the case presentation.
5. List of Abbreviations used in the manuscript has been provided.
6. Figure legends have been revised/elaborated to match with the figures.
7. Confidentiality of the patient has already been ensured.

Please do let us know if there are any further requirements.

Best regards,

Dr. Abdul Mueed Zafar
Aga Khan University,
Karachi, Pakistan.